Author's response to reviews

Title: Population screening for hereditary and familial cancer syndromes in Valka district of Latvia

Authors:

Andrejs Vanags (vanags314@inbox.lv)
Ilze Štrumfa (ilze.strumfa@rsu.lv)
Andris Gardovskis (andris.gardovskis@stradini.lv)
Viktors Boro&353;enko (vborosenko@stradini.lv)
Arnis &256;bol&326; (&353; arnis.abolinsh@inbox.lv)
Uldis Teibe (uldis.teibe@rsu.lv)
Genadijs Trofimovs (trofimovic@navigator.lv)
Edv&299;ns Mikla&353;evi&269; (edvins.miklasevics@stradini.lv)
J&257;nis Gardovskis (janis1105@inbox.lv)

Version: 3 Date: 20 October 2010

Author's response to reviews: see over
Dear Sir,

According to the Recommendations, we have made the following formatting changes in the manuscript “Population screening for hereditary and familial cancer syndromes in Valka district of Latvia” by A. Vanags, I. Štrumfa, A. Gardovskis, V. Borošenko, A. Ābolinš, U. Teibe, G. Trofimovičs, E. Miklaševičs, J. Gardovskis:

1. In the section Authors’ contributions (page 14) we have indicated that all authors read and approved the final manuscript.
2. The tables are included in the manuscript following the references. Any visible vertical lines have been removed from the tables.
3. The references in the text have been formatted appropriately (page 12).
4. The phrase „Research article” has been removed from the title page.
5. In the section References, the list of authors has been provided in the entirety.
6. The manuscript has been checked for any typographical errors.

Thus, all recommended major and minor revisions have been implemented.

On behalf of all authors, thank You for the support.

Yours sincerely,
Andrejs Vanags